Author's response to reviews

Title: Higher Plasma Homocysteine in Non-Alcoholic Fatty Liver Disease (NAFLD)

Authors:

Sylene Carvalho (sylenerampche@hotmail.com)
Maria Tereza Muniz (tcartaxo.upe@gmail.com)
Maria Deozete Siqueira (deta.siqueira@yahoo.com.br)
Érika Siqueira (erikarfs@hotmail.com)
Adriana Gomes (adrianavieiragomes@gmail.com)
Karina Silva (Karina_upe@yahoo.com.br)
Lais Bezerra (lais_luma@hotmail.com)
Vania Almeida (vaniadalmeida@uol.com.br)
Claudia Oliveira (cpm@usp.br)
Leila Pereira (leilapereira@pq.cnpq.br)

Version: 3 Date: 28 January 2013

Author's response to reviews: see over
Nutrition Journal

I would like to resubmit the paper titled **Plasmatic Higher Levels of Homocysteine in Non-Alcoholic Fatty Liver Disease (NAFLD)** to this journal, in view of the request of Dr. Hiromichi Kumagai.

I consider that this research is related with the scope of this journal and I hope that this study bring a contribution on Fatty Disease research. The authors declare that they do not have competing interests.

Sincerely,

Profª. Drª. Maria Tereza Cartaxo Muniz
Instituto de Ciências Biológicas, Universidade de Pernambuco.